Author’s response to reviews

Title: Clinical and genetic analysis of two wolfram syndrome families with high occurrence of wolfram syndrome and Diabetes type II: a case report

Authors:

Maryam Sobhani (msobhani@alumnus.tums.ac.ir)
Mohammad Amin Tabatabaiefar (mamintab@yahoo.co.uk)
Soudeh Ghafouri-Fard (s.ghafourifard@sbmu.ac.ir)
Asadollah Rajab (asadrajab@yahoo.com)
Asal Hojat (asalhojat@gmail.com)
Abdol-Mohammad Kajbafzadeh (kajbafzd@tums.ac.ir)
Mohammad Reza Noori-Daloii (nooridaloii@sina.tums.ac.ir)

Version: 2 Date: 26 Oct 2019

Author’s response to reviews:

Response to comments

1. We notice significant text overlap between your manuscript and previously published materials. Please reformulate the large sections of overlapping text present in your Background, clinical examinations and sampling and DNA extraction sections. Please ensure that, where relevant, these sources are also referenced as appropriate

“MUTATION ANALYSIS OF WOLFRAM SYNDROME PATIENTS AND FUNCTIONAL STUDY OF THE WOLFRAMIN PROTEIN”
https://etheses.bham.ac.uk/id/eprint/3760/1/Prince12PhD.pdf

“Molecular characterization of WFS1 in an Iranian family with Wolfram syndrome reveals a novel frameshift mutation associated with early symptoms”

Response: We reformulated the text and cited the mentioned references.

2. On the title page of your manuscript please indicate only one corresponding author. Please ensure that the sole corresponding author on the title page is consistent with the corresponding author declared in the submission data.
Response: We indicated the corresponding author.
3. In your Ethics approval section, please clarify whether consent obtained from participants was written or verbal. If consent was obtained verbally please include reasoning for this and state if an IRB approved this method of obtaining consent.
Response: We indicated the note.

4. As you report several individual cases in your manuscript, and for all manuscripts that include details relating to an individual person, written informed consent for the publication of these details must be obtained from that person (or their parent or legal guardian in the case of children under 18). Please provide the Consent for Publication as the following format if you have obtained the Consent for Publication in written form.

“Written informed consent was obtained from the participants for publication of their cases and any accompanying tables/images. A copy of the written consent is available for review by the Editor of this journal.”
Response: We indicated the mentioned point.

5. All sources of funding for the research reported should be declared in this section. Please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: We mentioned the point.

6. Please provide more justification for the contributions of authors. We ask that you consider your list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors MAT, AMK and MRND do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):
Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Response: We added further clarifications on the contributions of the mentioned authors.

7. Please remove the CARE checklist.
Response: We removed the CARE checklist.

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: We uploaded the file as stated.